Author's response to reviews

Title: Promoting menstrual health among Persian adolescent girls from low socioeconomic backgrounds: A quasi-experimental study

Authors:

Moloud Fakhri (mmfir@yahoo.com)
Zeinab Hamzehgardeshi (Hamzeh@razi.tums.ac.ir)
Nayereh Azam Hajikhani Golchin (Hajikhani2@yahoo.com)
Abdulhay Komili (Abdolhaykomili@mail.ru)

Version: 6 Date: 23 February 2012

Author's response to reviews: see over
RESPONSE TO REVIEWER COMMENTS AND SUGGESTION

Manuscript Title: MS: 6207150196064275 - Promoting menstrual health among Persian adolescent girls from low socioeconomic backgrounds: A quasi-experimental study

We thank the reviewers for their careful read and re-review on previous revised draft. Reviewer’s report was “The authors have answered the points sufficiently well. The revised manuscript is acceptable“.
Thank you for your advice that we pay close attention to language during revision of this manuscript. The first revised version of this manuscript was edited by Edanz.

Thanks to reviewers for the thoughtful and thorough re-review. Hopefully we have addressed all of your concerns.